Author's response to reviews

Title: Dupuytren's contracture: a retrospective database analysis to assess clinical management and costs in England

Authors:

Robert A Gerber (robert.a.gerber@pfizer.com)
Richard Perry (richard.perry@adelphigroup.com)
Robin Thompson (Robin.Thompson@opml.co.uk)
Chris Bainbridge (chris@hand-surgery.co.uk)

Version: 8 Date: 11 April 2011

Author's response to reviews:

Two minor revisions needed:
1. Paragraph in results section just above the title 'Discussion': please remove hyphen for Day-case in second sentence.
2. In Discussion: 4th paragraph, sentence beginning with 'In a 6-month study of 76..., please remove hyphen in day-case.